Author’s response to reviews

Title: The effect of motivational interviewing on oral healthcare knowledge, attitudes and behaviour of parents and caregivers of preschool children compared to traditional dental health education: A feasibility study

Authors:

Rahul S Naidu (rsnaidu937@gmail.com)
June H Nunn (june.nunn@dental.tcd.ie)
Jennifer D Irwin (jenirwin@uwo.ca)

Version: 5  Date: 9 March 2015

Author's response to reviews:

Dear Editor,

As requested our manuscript has been revised with the following changes:

1/ No mark up.
2/ Table wrapping removed.
3/ Figure titles removed.
4/ No boxes around figure legends.

King regards,
Dr. Rahul Naidu